0. Surv ey Results

1

1. Ev er had a CG class b efore?

YES 11

NO 9

F or those who ha v e had a course, it w as b et w een 1979

and \a few y ears ago," with the median y ear b eing

around 1990.

The courses w ere at UW (3), Stanford (2), BYU, Ohio

State, Seattle U, and Colorado State.

2

2. What do y ou hop e to get out of the class?

Answ ers w ere all o v er the place:

� \Basic kno wledge of 2D and 3D rendering and

animation"

� \More than clipping and 3D pip eline"

� \Op enGL and Direct3D"

� \API agnostic real-time 3D rendering tec hniques"

� \T ric ks in 3D games and animation"

� \Data visualization"

� \Curren t dev elopmen ts in the �eld"

3

3. What kind of w ork do y ou do?

Again, lots of di�eren t things:

� Application dev elopmen t

� Streaming net w ork video/audio

� Sim ulations of aircraft and a vionics

� Engineer on t yp ograph y related issues

� W eb stu�

� Natural languages

4

4. Preferred classro om exp erience?

Lectures 8

Studen t presen tations 0 Com bination 12

Those w an ting just lectures w ere adaman t. Those w an ting com bination still w an ted primarily lectures. 5

5. What kind of assignmen ts? Programming 16

Homew orks 7

Reading and presen ting 8

Artifacts 11 Com bination 15

Most w an ted small programming pro jects.

Some v oiced concern o v er partner-based pro jects. 6 6. Ho w man y hours/w eek outside of class?

Ranged from \3{4" to \as m uc h as it tak es."

Median is ab out 10 hours.

7

7. T opics, sorted b y lev el of in terest

Graphics programming 137

Rendering tric ks 132

Curv es and surfaces 128

Hidden surface 124

Image pro cessing 119 Ra y tracing 117 Lev el-of-detail 116

Shading 115

Morphing 111

Geometric transformations 106

Visual p erception 104

Hierarc hical mo deling/animation 103 8

8. Other sub jects?

� VRML

� Graphics standards

� Image recognition

� Games programming

� Real-time and p erformance issues

� Progressiv e mesh rendering

� Query b y image con ten t � Image and video storage and pro cessing 9

9. F undamen tals vs. cutting edge F undamen tals 13

Cutting edge 9 Man y ask ed for b oth: fundamen tals, then latest dev elopmen ts. 10 10. Breadth vs. depth

Breadth 14

Depth 6

11